Author's response to reviews

Title: Prevalence and barriers to HIV testing among mothers at a tertiary care hospital in Phnom Penh, Cambodia

Authors:

Yuri Sasaki (yurippe1006@hotmail.com)
Moazzam Ali (denube9@yahoo.com)
Vong Sathiarany (sathiarany@yahoo.com)
Koum Kanal (koumkanal@camnet.com.kh)
Kazuhiro Kakimoto (k-kakimoto@it.ncgm.go.jp)

Version: 5 Date: 15 August 2010

Author's response to reviews: see over
To 
The Editor, 
BMC Public Health 

Subject: Resubmission of article title; “Prevalence and barriers to HIV testing among mothers at a tertiary care hospital in Phnom Penh, Cambodia” 

Dear Sir or Madam

We are resubmitting this article to your prestigious journal under the above-mentioned title. This is to declare that this manuscript is being submitted to only, that it has not been, and will not be submitted elsewhere, and that should it be published in BMC Public Health, it will not be published elsewhere without permission of the Editors.

We hope kind consideration will be given to it.

Best Regards

Corresponding Author,
Ms. Yuri Sasaki

School of International Health
Graduate School of Medicine
The University of Tokyo
7-3-1 Hongo, Bunkyo-Ku, Tokyo 113-0033, Japan.
Tel: +81- 3 - 5841- 3528
E-mail: yurippe1006@hotmail.com